Author’s response to reviews

Title: The effect of very brief smoking reduction intervention in smokers who have no intention to quit: study protocol for a randomized controlled trial

Authors:

Lei Wu (wlyg0118@163.com)
Yao He (yhe301@263.net)
Bin Jiang (jiangbin301@sina.com)
Di Zhang (zdenjoylife@126.com)
Hui Tian (epid301@163.com)
Fang Zuo (annzuof@shou.com)

Version: 4 Date: 8 March 2015

Author’s response to reviews: see over
Dear editor,

We would like to submit the enclosed manuscript “The effect of very brief smoking reduction intervention in smokers who have no intention to quit: study protocol for a randomized controlled trial” for consideration of publication in *BMC public health*.

We are most grateful for the comments from the editor. We have made the substantial changes accordingly in this new article as follows.

1. We confirm that we have forwarded the requested Ethical and Funding Approval Documentation to BMCSeriesEditorial@biomedcentral.com.
2. We confirm that the study protocol has not undergone peer-review by the funding body.
3. The project is ongoing, and the investigators are still collecting.
4. We confirm that publications containing the results of this study have not been published or submitted to any journal.
5. We have changed the improper expression and improved the writing.
6. We have provided the Trial Registration Number (TRN) at the end of the Abstract in the submission system.

Thanks for your consideration.

Your regards,

Lei Wu, MPH

Department of Epidemiology, Institute of Geriatrics, Chinese PLA General Hospital, 28 Fuxing Road, Beijing 100853, China
Yao He, MD, PhD,

Professor of Epidemiology

Department of Epidemiology, Institute of Geriatrics, Chinese PLA General Hospital, 28 Fuxing Road, Beijing 100853, China